Author's response to reviews

**Title:** Sub-acute dislocation of elbow following Galeazzi fracture dislocation of radius: A case report

**Authors:**

aysha a rajeev (asrajeev18@gmail.com)
shanaka s senevirathna (shanaka_s2004@yahoo.com)
sarkell s radha (sarkell.radha@ghnt.nhs.uk)
john j harrison (john.harrison@ghnt.nhs.uk)

**Version:** 2  **Date:** 12 July 2011

**Author's response to reviews:**

Dear Editor,

As requested we have made all the necessary corrections in the manuscript

1. Title page
2. New abstract
3. Discussion and conclusion separated
4. Consent section included
5. Competing interest and author contributions added
6. Figure and legend list
7. All the changes in the manuscript highlighted

Thankyou

S Senevirathna